Author’s response to reviews

Title: Disparate compensation policies for research related injury in an era of multinational trials: A case study of Brazil, Russia, India, China and South Africa.

Authors:

Rugare Chingarande (rgc36d@mail.missouri.edu)

Keymanthri Moodley (km@sun.ac.za)

Version: 2 Date: 30 Nov 2017

Author’s response to reviews:

1. Title page has been provided.

2. Abstract, keywords have been included.

3. Methodology replaced with Methods.

4. Not applicable removed from Ethics statement

5. The other ethics section has been removed.

6. Availability of data statement has been updated to indicate that the data is available on request.

7. Track changes has been deactivated and document cleaned up.